CENWP-OP-B                                                                                                                                           06 July 2003

MEMORANDUM FOR CENWP-OP, ATTN: Gary Johnson

SUBJECT: Bonneville Lock and Dam, Fishway and Fish Activities for week 27 of 2003, which covers the period 29 June through 05 July 2003.

1.  OPERATION SUMMARY.

1.1 Daily river flow averaged 144.6 to 205.3 kcfs.  Daily project tailwater elevation averaged 11.8’ to 16.1’ msl.  Daily project forebay elevation averaged 74.7’ to 75.5’ msl.  River temperature averaged 62 to 63° F.  Daily turbidity measured 4.8’ to 5.0’ on the secchi disk.

1.2 Spill Bays 1 and 18, which supply attraction flow for Cascade Island and Bradford Island fishways respectively, are included in the spill pattern.  Spill averaged 83.4 to 94.9 kcfs.  

1.3 Unit Operation

1.3.1 Unit 1 remains out of service for turbine rehab.

1.4 Second Powerhouse fish units provide attraction flow for the Washington Shore fish ladder.  The units are placed on standby from approximately 0000-0300 nightly if drawdown exceeds 1.0’.   

1.5 Gatewell drawdowns are measured on Wednesdays.

2.  ACTIVITIES. 
2.1 AWS Closures.  The following AWS valves were closed this reporting week.

FV3-7 closed for cleaning from 1007 until 1056 on 02 July.

FV3-9 closed for cleaning from 1255 until 1306 on 30 June.

Washington Shore exit and FV6-9 throttled for debris removal from 0759 until 0847 on 01 July.

FV1-1, FV3-7, FV4-3, FV4-4 in manual on 02 July.  See misc. 5.6. 
2.2 STS/VBS Inspections.  The following STS/VBS inspections were conducted this reporting week.

	Unit
	Date
	Run Hours
	Comments

	11
	7/1
	664
	

	18
	7/1
	673
	


Table 1.  STS Inspection Dates.

2.3 Dewatering and Fish Salvage.  No dewatering or fish salvage activities this reporting week.

3.  RESEARCH.

3.1 CRITFC.  Sampling of chinook, sockeye, and coho continues until 31 October.

3.2 NMFS Lamprey.  Trapping, tagging and tracking continue.

3.3 OCFWRU.  Fish collection continues. 

3.4 ODFW Steelhead.  Trapping continues. 

3.5 PSMFC at DSM1.  Continuous PIT tag interrogation continues.  

3.6 PSMFC at the SMF.  Four steelhead, three salmon and 15 smolts slid over the separator bars.  GBT exams continue.

3.7 University of Idaho.  Adult tracking tagging and tracking continue.  

3.8 U of I Transportation Study.  Evaluation of river run vs. barged Snake River juveniles to occur May-June.   Barge sampling concluded on 26 June.  Will be periodically sampling fall Chinook to 15 August.

3.9 University of Washington.  Northern pikeminnow tagging and tracking from May through July.  

3.10 WDFW Fish Counting.  16-hr visual counts at Washington Shore and Bradford Island.

3.11 WES/Battelle/BAE systems Juvenile hydroacoustic monitoring.  Gap loss monitoring continues.

4.  FISHWAYS 
4.1 Project biologists inspected 29 June through 05 July this reporting week.

4.2 Fish Passage Plan Observations.

	DATE
	LOCATion
	FPP Deviation
	CAUSE
	RESPONSE
	TR/WAR

	All week
	PH1 PCC
	Velocity <1.5fps
	Meter in slowest part of channel
	No response at this time.
	

	All week
	PH2 PCC
	No velocity measure.
	Meter is missing
	See misc. 5.1
	

	6/30, 7/4
	Weir 37 
	Reading >or <1.3’ +/- .1’
	Change in FB elev.
	
	

	6/30
	N and S monoliths
	Head differential <1.0’
	Change in tailwater elevation.
	Operators to adjust gates.
	

	All week
	Bonneville Project
	Fishway calibration not checked.
	
	
	


4.3 Fish Ladder Temperatures.  Measurements are taken three weirs down from the count station at Bradford Island and at the junction of the UMT and the main ladder at Washington shore.  Temperatures are in Kelvin.  http://students.washington.edu/kyle/temptable.jpg 

	Date
	Wash. Shore
	Bradford Is.
	AFF exit

	6/29
	291.1
	291.2
	291.1

	6/30
	291.2
	291.3
	291.2

	7/1
	291.2
	291.2
	291.2

	7/2
	291.2
	291.2
	291.2

	7/3
	291.1
	291.2
	291.1

	7/4
	291.3
	291.3
	291.3

	7/5
	291.5
	291.5
	291.5


4.4 Sensor Calibration.  Calibration was not checked this reporting week.

4.5  Zebra Mussel Monitoring.  No signs of colonization were observed this reporting week. 

5. MISCELLANEOUS

5.1 Divers discovered the PH2 velocity meter was missing from the collection channel.  A new meter will be installed during the mid season dives scheduled for 21 and 22 July.

5.2 As of 1500 on 30 May, the Project is restricted to a maximum tailwater of 27’.  This is to prevent overtopping the bulkhead in the B2 Corner Collector.  The bulkhead went in about a month ago and is located at Station 24 +56 (near the start of the monolith).  

5.3 DSM2 Screen Cleaner #1 was out of service from 1230 until 1530 on 27 July.

5.4 Orifice 18A-N remains in local and open due to a bad proximity switch.

5.5 As of 28 June, the Project will have a daytime spill of 75 kcfs.

5.6 District Engineers were out taking measurements for the BICI evaluation on 02 July.  At their request FV1-1 was in manual from 1400 until 1653, FV3-7 was in manual from 1400 until 2000, FV4-3 and FV4-4 were in manual from 0815 until 1350.

5.7 AFF picket lead #2 went out of service at 0925 on 02 July.  The hoist would not raise the lead.  Mechanics have removed the hoist leaving the picket lead in a permanently closed position until repairs are completed, which should be early next week.

JAMES R. MAHAR, P.E.

Operations Manager
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